Author's response to reviews

Title: Internet-based treatment for older adults with depression and co-morbid cardiovascular disease: protocol for a randomised, double-blind, placebo controlled trial

Authors:

Nicole L Cockayne (nicole.cockayne@sydney.edu.au)
Nick Glozier (nick.glozier@sydney.edu.au)
Sharon L Naismith (sharon.naismith@sydney.edu.au)
Helen Christensen (helen.christensen@anu.edu.au)
Bruce Neal (bneal@george.org.au)
Ian B Hickie (ian.hickie@sydney.edu.au)

Version: 2 Date: 4 December 2010

Author's response to reviews:

Dear Editor,

Upon review of the manuscript following submission we noted an error in the description of the tests used to measure cognitive function. Specifically, this error related to the first cognitive test, which is a test of psychomotor functioning as opposed to visual attention. The manuscript has since been updated to reflect this (refer to page 15).

We would like to apologise for the oversight and trust that this will not adversely affect our opportunity to publish the protocol in your journal.

Yours sincerely,

Nicole Cockayne